Author's response to reviews

Title: A Population-based study of dementia in the oldest old: the Monzino 80-plus Study Design, methodological challenges, and population characteristics

Authors:

Ugo Lucca (lucca@marionegri.it)
Mariateresa Garri (mariateresa.garri@marionegri.it)
Angela Recchia (angela.recchia@marionegri.it)
Giancarlo Logroscino (giancarlo.logroscino@neurol.uniba.it)
Pietro Tiraboschi (ptiraboschi@yahoo.com)
Massimo Franceschi (massimo.franceschi@multimedica.it)
Chiara Bertinotti (chiarabertinotti@libero.it)
Anna Biotti (anna.biotti@libero.it)
Elena Gargantini (elena.gargantini@tin.it)
Marilena Maragna (malinenam.m@libero.it)
Alessandro Nobili (alessandro.nobili@marionegri.it)
Luca Pasina (luca.pasina@marionegri.it)
Carlotta Franchi (carlotta.franchi@marionegri.it)
Emma Riva (emma.riva@marionegri.it)
Mauro Tettamanti (mauro.tettamanti@marionegri.it)

Version: 4 Date: 20 May 2011

Author's response to reviews:

To the attention of
The BMC Series Editorial Production Team

We have made all the requested major and minor revisions.
We have checked carefully but were unable to find any difference between the title and subtitle in the manuscript and that in the submission system:

A Population-based study of dementia in the oldest old: the Monzino 80-plus Study
Design, methodological challenges, and population characteristics

Best wishes